Author's response to reviews

Title: Leptospirosis in the Asia Pacific Region

Authors:

Anne Florence B Victoriano (entyv@yahoo.com)  
Lee D Smythe (Lee_Smythe@health.qld.gov.au)  
Nina Gloriani-Barzaga (ninagloriani@yahoo.com)  
Lolita L Cavinta (cavinta@yahoo.com)  
Takeshi Kasai (kasait@wpro.who.int)  
Khanchit Limpakarnjanarat (khanchitl@searo.who.int)  
Bee Lee Ong (ongb@wpro.who.int)  
Gyanendra Gongal (Gongalg@searo.who.int)  
Julie Hall (hallj@wpro.who.int)  
Caroline Anne Coulombe (coulombec@wpro.who.int)  
Yasutake Yanagihara (yanaghira@uv.tnc.ne.jp)  
Shin-ichi Yoshida (syn-ichi@bact.med.kyushu-u.ac.jp)  
Ben Adler (Ben.Adler@med.monash.edu.au)

Version: 4  Date: 24 August 2009

Author's response to reviews:

Attached is the final version with corrections as requested by the editors. Please note that one of the important features of our review is the fact that we have accessed information that on occasion may not fit with the classical reference format, so it’s not possible to quote the reference in the normal format. We have tried to provide as much information as possible to detail the source.

sincerely,

Ben Adler